Author's response to reviews

Title: No long-term impact of low-energy distal radius fracture on health-related quality of life and global quality of life. A case-control study.

Authors:

Gudrun Rohde (gudrun.rohde@sshf.no)
Glenn Haugeberg (glenn.haugeberg@sshf.no)
Anne Marit Mengshoel (a.m.mengshoel@medisin.uio.no)
Torbjorn Moum (torbjorn.moum@medisin.uio.no)
Astrid K Wahl (a.k.wahl@medisin.uio.no)

Version: 23 Date: 17 August 2009

Author's response to reviews:

The requested changes have been made